Author’s response to reviews

Title: Effect of a multi-ingredient based food supplement on sexual function in women with low sexual desire. Pilot study.

Authors:

Santiago Palacios (ipalacios@institutopalacios.com)

Eva Maria Soler (eva.maria.soler@institutopalacios.com)

MARIETA RAMIREZ (marieta.ramirez@institutopalacios.com)

MARIELLA LILUE (mariella.lilue@institutopalacios.com)

Danial Khorsandi (khorsandi.d@procarehealth.com)

FERNANDO LOSA (losa.fernando@gmail.com)

Version: 6 Date: 08 Apr 2019

Author’s response to reviews:

1. Due to the design of your study - small sample size, no control group, in the Discussion section please expand further on the limitations of this study, and how these may affect the interpretation of the results.

Included in the Discussion section, in red.

2. If available, please provide your trial registration number in the last section of the abstract - Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section.

Already included at the end of the Abstract, as requested, also in red.